Author's response to reviews

Title: TLR9 expression in glioma tissues correlated to glioma progression and the prognosis of GBM patients

Authors:

- Chao Wang (n_wangchao@yahoo.com)
- Shouqiang Cao (csqrock@tom.com)
- Ying Yan (yanying_doctor@hotmail.com)
- Ying Qiao (gfm1932@yahoo.com.cn)
- Tao Jiang (tiantan_jiangtao@yahoo.com)
- Ke Xu (xuke_cmu1h@hotmail.com)
- Anhua Wu (anhuawu@yahoo.com)

Version: 5 Date: 9 August 2010

Author's response to reviews: see over
Dear Editors,

The enclosed is my revised manuscript entitled ‘TLR9 expression in glioma tissues correlated to glioma progression and the prognosis of GBM patients’, with MS ID: 1820358524329433. The format of the manuscript was revised according to the editor’s request. point-by-point response to editor’s request is appended below.

Correspondence should be addressed to Wu Anhua at the following address, phone and fax number, and email address:

Wu Anhua  
Department of Neurosurgery  
First Hospital of China Medical University  
No.155, North Nanjing Street, Heping District  
Shenyang, Liaoning, P.R.China,110001  
Tel:86-24-23517483  
Fax:86-24-23517819  
E-mail:anhuawu@yahoo.com

Sincerely yours,  
Wu anhua
Major revisions (we require the author to make these changes)

1. Figure 2: Please provide a short overall title for figure 2. Each figure should have a single title - figures can have multiple parts and corresponding legends, but should have a single composite/overall title.

   Our revision: a short overall title has been provided for figure 2.